

Impact of single nucleotide polymorphisms on the efficacy and toxicity of egfr tyrosine kinase inhibitors in advanced non-small cell lung cancer patients.

Perez-Ramirez, C., Cañadas-Garre, M., Molina, M. A., Cabeza Barrera, J., & Faus Dader, M. J. (2019). Impact of single nucleotide polymorphisms on the efficacy and toxicity of egfr tyrosine kinase inhibitors in advanced non-small cell lung cancer patients. *Mutation research-Reviews in mutation research*. Advance online publication. https://doi.org/10.1016/j.mrrev.2019.04.001

Published in:

Mutation research-Reviews in mutation research

Document Version:

Peer reviewed version

Queen's University Belfast - Research Portal:

Link to publication record in Queen's University Belfast Research Portal

Publisher rights

Copyright 2019 Elsevier.
This manuscript is distributed under a Creative Commons Attribution-NonCommercial-NoDerivs License (https://creativecommons.org/licenses/by-nc-nd/4.0/), which permits distribution and reproduction for non-commercial purposes, provided the author and source are cited.

General rights

Copyright for the publications made accessible via the Queen's University Belfast Research Portal is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy
The Research Portal is Queen's institutional repository that provides access to Queen's research output. Every effort has been made to ensure that content in the Research Portal does not infringe any person's rights, or applicable UK laws. If you discover content in the Research Portal that you believe breaches copyright or violates any law, please contact openaccess@qub.ac.uk.

Open Access

This research has been made openly available by Queen's academics and its Open Research team. We would love to hear how access to this research benefits you. - Share your feedback with us: http://go.qub.ac.uk/oa-feedback

Download date: 18 Jul 2024

INDEX

1	TIT	LE	
2	ΛD	STRACT	,
_			
3	INT	RODUCTION	4
4 EGFR PATHWAY			5
5	AK ⁻	T PATHWAY	6
6	TGI	F-B PATHWAY	6
7	CELLULAR EFFLUX TRANSPORTERS		7
	7.1	ABCB1	7
	7.2	ABCG2	7
8	CYT	FOCHROME P450 FAMILY	
	8.1	CYP3A4/5	
	8.2	CYP2D6	
	8.3	CYP1A1	8
9	CO	NCLUSIONS	8
1(O REF	ERENCES	9

1 TITLE

IMPACT OF SINGLE NUCLEOTIDE POLYMORPHISMS ON THE EFFICACY AND TOXICITY OF EGFR TYROSINE KINASE INHIBITORS IN ADVANCED NON-SMALL CELL LUNG CANCER PATIENTS.

AUTHORS:

Cristina Pérez-Ramírez^{1,2}, Marisa Cañadas-Garre³, Miguel Ángel Molina⁴, José Cabeza Barrera¹, María José Faus-Dáder².

Author for correspondence:

3 Marisa Cañadas-Garre, Ph.D.

Queen's University Belfast Centre for Public Health Nephrology Research Group

c/o Regional Genetics Centre, Level A, Tower Block

Belfast City Hospital, Lisburn Road

Belfast, BT9 7AB

Telephone: +44 02890 638460

Fax: +44 02890 235900

E-mail: marisacgarre@gmail.com

Affiliations

1 Pharmacogenetics Unit

UGC Provincial de Farmacia de Granada Instituto de Investigación Biosanitaria de Granada Complejo Hospitalario Universitario de Granada

Avda. Fuerzas Armadas, 2
Telephone: +34958020108
Fax: +34901021804

2 Department of Biochemistry

Faculty of Pharmacy
University of Granada
Campus Universitatio de Cart

Campus Universitario de Cartuja, s/n

18071 Granada, Spain

Telephone: +34958243838

4 Pangaea Biotech, S.L.

Hospital Universitario Quirón Dexeus

C/ Sabino Arana, 5-19. 08028 Barcelona. SPAIN

Tel.: +34 93 546 01 35 Fax: +34 93 546 01 72

Mails

Cristina Pérez-Ramírez cperezramirez87@gmail.com
Marisa Cañadas-Garre marisacgarre@gmail.com
Miguel Ángel Molina mamolina@panoncology.com

José Cabeza Barrera jose.cabeza.sspa@juntadeandalucia.es

María José Faus-Dáder mfaus@ugr.es

Keywords

EGFR TYROSINE KINASE INHIBITORS NON-SMALL CELL LUNG CANCER RESPONSE SURVIVAL TOXICITY POLYMORPHISMS

WORD COUNT (excluding references and figure legends):

Abstract: 234 words

Main Text: 2617 words

FUNDING

This work was partly supported by a research grant for Cristina Pérez-Ramírez (FPU12/04722), from Ministerio de Educación, Cultura y Deporte.

ACKNOWLEDGMENTS

The results of this investigation are part of the doctoral thesis presented by Cristina Pérez-Ramírez at the University of Granada.

2 ABSTRACT

EGFR tyrosine kinase inhibitors (EGFR-TKIs) are the treatment of choice for advanced-stage (IIIB-IV) NSCLC patients with mutations in *EGFR*. However, EGFR-TKIs clinical outcomes vary from person to person and these inter-individual differences may be due to genetic factors such as single nucleotide polymorphisms (SNPs). SNPs in genes involved in in EGFR-TKIs pharmacodynamics, metabolism and mechanism of action have been demonstrated to be associated with response, survival and toxicity in advanced NSCLC patients treated with EGFR-TKIs.

Here we review the influence of gene polymorphisms in the EGFR pathway on clinical outcome and toxicity to EGFR-TKIs in advanced NSCLC patients. The *EGFR*-216 polymorphism has reported a strong association between response and/or survival to EGFR-TKIs in Caucasian population. Similarly, the effect of EGFR-CA repeats polymorphisms on survival of advanced NSCLC patients treated with EGFR-TKIs have been confirmed both in Caucasian and Asian population. The influence on toxicity of the -216, -191, CA repeats, Arg497Lys and Asp994Asp polymorphisms in *EGFR* have also been confirmed. Polymorphisms in *AKT* (rs1130214 and rs1130233) and *SMAD3* (rs6494633, rs11071938 and rs11632964) have been associated with survival in advanced NSCLC patients treated with EGFR-TKIs. However, data come from a limited number of studies and need to be confirmed.

Finally, polymorphisms in genes coding proteins of the membrane transporters and cytochrome P450 enzymes have been less extensively investigated. There are few studies with small samples, which complicated the generalization of their role in EGFR-TKIs treatment.

3 INTRODUCTION

Lung cancer is one of the most common and lethal types of cancer in both genders, with an approximate incidence of 14% [1]. Based on the latest cancer statistics, around 222.500 new cases (116.990 in male and 105.510 in female) and 155.870 deaths (84.590 in male and 71.280 in female) are expected to occur in the United States in 2017 [1].

There are two main types of lung cancer: small cell (SCLC) and non-small cell lung cancer (NSCLC). NSCLC accounts with 80-85% of all lung cancer cases and is classified into three different subtypes: squamous cell carcinoma, adenocarcinoma and large cell carcinoma. In accordance with the American Joint Committee on Cancer (AJCC), the majority of the patients are catalogued as advanced stage (IIIB-IV) at the time of diagnosis [2-4].

For many years, platinum-based chemotherapy has been the treatment of choice for advancedstage (IIIB-IV) NSCLC [5]. Nevertheless, targeted therapy has emerged as a therapeutic option for selected patients. Patients with somatic, activating mutations in EGFR (epidermal growth factor receptor) are treated with an EGFR tyrosine kinase inhibitor (EGFR-TKI), such as gefitinib or (Iressa®; AstraZeneca, London, UK), erlotinib (Tarceva®; Hoffmann-La Roche, Basel, Switzerland), afatinib (Giotrif®; Boehringer Ingelheim, Ingelheim, Germany) or osimertinib (Tagrisso ®; AstraZeneca, London, UK) [6-10]. Activation of the EGFR pathway is induced by ligand binding, which results in receptor dimerization and phosphorylation of tyrosine residues located at the cytoplasmic tail of the receptor, leading to phosphorylation of effector proteins [11,12] (Figure 1). Subsequently, downstream cascades, including the anti-apoptotic Ras signal transduction cascade (KRAS-BRAF-MEK-ERK pathway), the phosphatase phosphatidylinositol 3-kinase / tensin homolog /v-akt murine thymoma viral oncogene (PI3K/PTEN/AKT), phospholipase C gamma protein pathway, and the STAT signaling pathway are activated, leading to cell proliferation, angiogenesis, migration, survival, and adhesion [13] (Figure 1). EGFR-TKIs are orally active compounds that act by binding to the adenosine triphosphate (ATP)-binding domain of EGFR. The inhibition of the receptor leads to a blockade of downstream cascades, which induce cancer cell death in EGFR mutated cancer cells [14]. There are two type of EGFR-TKIs that differ in their abilities to fit in the ATP-binding pocket of EGFR. First generation or reversible inhibitors, such as gefitinib and erlotinib, compete with ATP molecules that recognize the kinase active conformation, whereas second generation or irreversible inhibitors such as afatinib, bind to the kinase active site covalently by specifically reacting with a nucleophilic cysteine residue [15]. Third generation inhibitors, such as osimertinib, are irreversible EGFR-TKIs selective for both *EGFR* sensitizing mutations and *EGFR* Thr790Met resistance mutation [16].

Activating mutations in the *EGFR* gene appear more frequently in adenocarcinoma subtype, females, non-smokers and Asians [17-20]. The most frequent mutations in EGFR are small inframe deletions in exon 19 and a point mutation that replaces an arginine with a leucine at codon 858 (L858R) of exon 21 [21]. Several studies have compared first line EGFR-TKIs versus standard chemotherapy in patients with EGFR mutation-positive tumors, showing longer progression-free survival (PFS) (9.7 months vs 5.2 months), higher overall response rate (ORR) (71.2% vs 47.3%), a more favorable toxicity profile (28.7% vs. 61.0%) and better quality of life (48.0% vs. 40.8%) [8,22]. However, numerous studies have reported significant inter-individual differences in clinical outcomes to EGFR-TKIs, which may be due to genetic factors such as single nucleotide polymorphisms (SNPs) in particular genes [23].

At this respect, the influence of some SNPs in the *EGFR* gene itself have been extensively investigated (Table 1). As described above, AKT pathway also plays an important function on cancer cell proliferation and survival has been reported that SNPs in this gene may dysregulate signaling, promote tumorigenesis and contribute to individual variation in the response and toxicity to EGFR-TKIs [24,25].Finally, other pathways and proteins are also involved in toxicity and response to EGFR-TKIs including, the transforming growth factor beta (TGF-β) pathway, drug transporters, and the cytochrome P450 family. Acting in an opposite way, the TGF-β signaling pathway exerts a robust antiproliferative function [26] and polymorphisms in the genes of pathway may have an effect in the development of toxicity and disease progression to EGFR-TKIs. Genetic alterations in ATP-binding cassette, sub-family B (MDR/TAP), member 1 (*ABCB1*, also called *MDR1*) and ATP binding cassette subfamily G member 2 (*ABCG2*) have also been suggested as predictive markers of clinical outcomes and toxicity to EGFR-TKIs [27]. Finally, EGFR-TKIs are metabolized by members of cytochrome P450 family, mainly by CYP3A4/5, CYP2D6 and CYP1A1. Therefore, SNPs in these genes may modulate enzymatic activities and consequently act as pharmacogenetics predictors of response and toxicity to EGFR-TKIs.

4 EGFR PATHWAY

The most investigated polymorphisms in *EGFR* are rs712829 ($G \rightarrow T$ substitution at -216 upstream from the initiator codon), rs712830 ($C \rightarrow A$ substitution at -191 upstream from the initiator codon) and rs11568315 (CA simple sequence repeat in intron 1). *EGFR*-216 and -191 polymorphisms have been described to modulate "in vitro" the expression of *EGFR* gene [28]. *EGFR*-216 is located in a SP1-binding site, a transcription factor required for *EGFR* expression, and the substitution of C to C at this position has been shown to increase the promoter activity by 30% and the *EGFR* mRNA expression by 40% [28]. *EGFR*-191, located four nucleotides upstream of one of six transcription initiation sites, also modulates promoter activity, but not to the same extent as the *EGFR*-216 [28]. In addition, the length of the CA repeat has shown an inverse correlation with the expression of *EGFR* mRNA [29].

The relationship between *EGFR* polymorphisms and clinical outcomes have been investigated extensively. The *EGFR*-216 polymorphism has showed better overall survival (OS), PFS and ORR for patients with T allele (Table 1). For EGFR-191 polymorphism no association with clinical outcomes have been found (Table 1). However, a study in 175 Caucasian stage IB-IV NSCLC

patients evaluated *EGFR* -216G/-191C haplotype (G-C; *EGFR**1) and reported that the absence of EGFR*1 was associated with significantly better OS (HR=0.54; 95%CI=0.32, 0.91; non-EGFR*1 vs EGFR*1) and PFS (HR=0.65; 95%CI=0.42, 0.99; non-*EGFR**1 vs *EGFR**1) [30]. Regarding *EGFR* rs11568315 polymorphism, advanced NSCLC patients with shorter intron 1 CA repeats (<16 CA) of the *EGFR* gene showed an improved response, OS and PFS (Table 1).

An association between *EGFR* polymorphisms and toxicity have also been found in several studies. The T allele for *EGFR*-216 has been associated with skin rash and diarrhea (Table 1) [31,32]. In the case of *EGFR*-191, the A allele has been associated with diarrhea in 80 NSCLC patients [32]. In contrast, no association with toxicity was found after evaluating *EGFR* -216G/-191C haplotype in 109 Caucasian stage IIIA-IV NSCLC patients [33]. For *EGFR* rs11568315 polymorphism, a study with 52 Asian stage IIIB-IV reported that those with longer intron 1 CA repeats (>16 CA) of the *EGFR* gene showed a decreased risk to develop skin rash [34].

Two additional polymorphisms in EGFR, rs11543848 (G→A nonsynonymous substitution at codon 497, exon 13, Arg→Lys, Arg497Lys) and rs2293347 (G→A synonymous substitution at codon 994, exon 25, Asp→Asp, Asp994Asp), have also been investigated [35]. The A allele for EGFR rs11543848 seems to decrease the activity of EGFR [36,37]. EGFR rs2293347 does not change amino acid sequence of the protein and, to date, the possible functionality of this genetic alteration has not been evaluated. Nevertheless, synonymous polymorphisms may affect mRNA stability, translational kinetics, and splicing, resulting in alteration of protein amount, structure or function [38]. Both polymorphisms have been correlated with clinical outcomes in NSCLC patients treated with EGFR-TKIs [25,33,39-43]. The A allele for EGFR Arg497Lys polymorphism has been associated with longer OS in 225 Asian stage I-IV NSCLC patients with positive lymph node metastasis and previous platinum-based chemotherapy (Log-rank test, p = 0.0072 and p= 0.0038, respectively) [39]. A correlation between EGFR Arg497Lys-A allele and lower skin toxicity has also been reported in 96 Caucasian stage IIIB-IV NSCLC patients [33]. In contrast, the GG genotype has been associated with higher diarrhea IN [25]. No association between EGFR Arg497Lys polymorphism and ORR has been found [25,40]. Regarding EGFR Asp994Asp polymorphism, its association with clinical outcome to EGFR-TKIs remains unclear (Table 1), with some studies reporting better ORR in patients carrying the A allele [41] and others in patients with the G allele [42]. The same contradictory results have been reported in the case of PFS and OS, with some studies finding an association of the GG genotype with a better outcome and others reaching opposite conclusions (Table 1) [41-43].

5 AKT PATHWAY

Three SNPs for AKT have been studied; namely $G \rightarrow T$,rs1130214; $A \rightarrow G$, rs1130233 and $C \rightarrow T$ rs3730350. A Caucasian study with 230 advanced NSCLC patients treated with erlotinib, gefitinib or icotinib reported that patients with AKT rs1130214-GG genotype had longer PFS than those with the GT and TT genotypes (HR=1.39; 95%Cl=0.92, 1.95 for TT vs GG)[24]. For AKT rs1130233, the AA genotype was associated with shorter PFS (p=0.04) and OS (p=0.007) in 96 advanced NSCLC patients treated with gefitinib [25]. No association has been found between AKT rs3730350 and clinical outcomes in 96 Caucasian stage IIIB-IV advanced NSCLC patients treated with EGFR-TKIs [25].

6 TGF-B PATHWAY

The TGF- β signaling may function both as a tumor suppressor and as a tumor promoter pathway in a context-dependent manner via acting on SMAD transcriptional regulators [44]. This behavior depends on cell type and clinical stage of the tumor [44].

Three polymorphisms in *SMAD3* (C \rightarrow T, rs6494633; C \rightarrow T, rs11071938 and C \rightarrow T, rs11632964) were found to be associated with survival in 106 Asian stage IIIB-IV EGFR mutated NSCLC patients treated with EGFR-TKIs [45]. The rs649446633-CC, rs11071938-CT and rs11632964-CT

genotypes were associated with better PFS (HR=0.55; $Cl_{95\%}$ =0.37, 1.00 for CC vs CT/TT; HR=1.75; $Cl_{95\%}$ =1.06, 2.89 for CC vs CT/TT and HR=3.01; $Cl_{95\%}$ =1.54, 5.86 for CC vs CT/TT, respectively) [45]. The CT genotype in the *SMAD3* rs11632964 polymorphism was also associated with longer OS (HR=2.38; $Cl_{95\%}$ =1.15, 4.94 for CC vs CT/TT) [45].

7 CELLULAR EFFLUX TRANSPORTERS

ABCB1 and ABCG2 are considered the main EGFR-TKIs efflux transporters [46,47]. Polymorphisms in these genes have been shown to alter protein expression and/or activity of these transporters [48-56]. Thus, ABCB1 and ABCG2 polymorphisms may modify the elimination of EGFR-TKIs from the body and as a result affect treatment outcome.

7.1 ABCB1

ABCB1 belongs to the ATP-binding cassette family and plays an essential function on efflux and distribution of many drugs, including EGFR-TKIs [57,58]. Polymorphisms in this gene have been associated with lower expression and function of the ABCB1 protein, resulting in increased extracellular levels of drugs [51-54]. Despite of this key role, none of the polymorphisms studied to date in the *ABCB1* gene (C→T, rs1045642; G→T/A, rs2032582; C→T, rs1128503) have shown a significant association with toxicity in NSCLC patients treated with EGFR-TKIs [27,59]. However, significant differences in toxicity have been demonstrated according to ABCB1 haplotype. A study with 50 Asian stage III-IV NSCLC patients treated with erlotinib have reported that the *ABCB1* rs1045642-TT; rs2032582-TT; rs1128503-TT haplotype was associated with higher plasma concentration of EGFR-TKI and the risk of developing higher toxicity [27]. The influence of these haplotypes on ORR, PFS and OS has not been determined.

7.2 ABCG2

ABCG2 is another member of the ATP-binding cassette family [60]. Genetic alterations in this gene have been has been associated with markedly decreased levels of ABCG2 protein expression and/or activity [48-50,55,56], which increases oral bioavailability of EGFR-TKIs [61]. A great variety of polymorphisms in ABCG2 gene have been studied such as $C \rightarrow T$, rs2622604; $C \rightarrow A$, rs2231142; $G \rightarrow A$, rs2231137, $G \rightarrow A$, rs7699188 and $C \rightarrow T$, rs72552713. Nevertheless, none of them have shown a significant association with clinical outcomes in NSCLC patients treated with EGFR-TKIs [62,63]. Only the A allele for rs2231137 has been correlated with grade 2 or worse skin rash in 83 Asian stage I-IV NSCLC patients treated with gefitinib (p=0.046) [59].

8 CYTOCHROME P450 FAMILY

EGFR-TKIs are metabolized in the liver by cytochrome P450 enzymes (CYPs), primarily by CYP3A4/5, CYP2D6 and CYP1A1 [64-67]. Polymorphisms in these genes may alter the metabolic activities of these enzymes and thereby drastically influence EGFR-TKIs plasma concentrations and detoxification, resulting in individual variation in response and toxicity to EGFR-TKIs [40,67-69].

8.1 CYP3A4/5

CYP3A4 and CYP3A5 are key enzymes for EGFR-TKIs metabolism [64-67]. To date, 34 CYP3A4 alleles (haplotypes) have been published on the Human Cytochrome P450 Allele Nomenclature Committee homepage [70]. However, their effects on outcome to EGFR-TKIs has not been investigated. Only the CYP3A4*1/*1G polymorphism ($G \rightarrow A$, rs2242480), within intron 10 of the CYP3A4 gene, has been studied in 31 Asian stage IIIB-IV NSCLC patients treated with gefitinib but no significant differences in toxicity was found [68]. For CYP3A5, 11 haplotypes have been described but only the CYP3A5*3 ($A \rightarrow G$, rs776746) polymorphism, within intron 3 of the CYP3A4 gene has been studied in 31 Asian stage IIIB-IV NSCLC patients treated with gefitinib [68]. Nevertheless, no significant association between CYP3A5*3 polymorphism and EGFR-TKIs was

found [68]. The relationship between both SNPs with response and survival has not been evaluated.

8.2 CYP2D6

CYP2D6 also plays a minor role on EGFR-TKIs metabolism [64-67]. A total of 109 CYP2D6 alleles have been described so far, but only the CYP2D6*1, *2, *3, *4, *5, *6, *9, *10 and *41 alleles have been studied. In 30 healthy volunteers treated with gefitinib [70], those with the CYPD6 extensive metabolizer genotype (*1/*4, *1/*2, *2/*4, *1/*3, *2/*5, *2/*41) presented higher gefitinib plasma concentration in comparison with those with CYPD6 poor metabolizer genotype (*4/*4, *4/*5, *3/*4, *4/*6, *3/*5, *4/*4x2) [67]. Two studies in Asian NSCLC patients treated with gefitinib have also evaluated the effect of CYPD26 (*5 and *10) polymorphisms on gefitinib toxicity but no significant differences were found in the frequency of diarrhea, skin rash, or hepatotoxicity among the genotypes of these polymorphisms [68,69]. Currently, no data are available regarding the influence of these SNPs on response and survival to EGFR-TKIs.

8.3 CYP1A1

CYP1A1 is a major enzyme involved in EGFR-TKIs metabolism [64-67]. Based on the CYP450 database, 13 CYP1A1 haplotypes has been described but only CYP1A1*2A ($T\rightarrow C$ substitution at 3' non-coding region) and CYP1A1*2C ($A\rightarrow G$ substitution at exon 7, Val \rightarrow Ile) have been examined in NSCLC patients treated with EGFR-TKIs [40]. Both CYP1A1*2A and CYP1A1*2C alleles have been associated with increased enzyme activity [71,72]. An Asian study with 115 advanced NSCLC patients treated with an EGFR-TKI reported that patients with CYP1A1*2A-TT had an improved response (p=0.011; TT vs CT/CC) and OS (HR=0.48; Cl_{95%}=0.31, 0.73 for TT vs CT/CC) to EGFR-TKI [40]. However, for CYP1A1*2C, no association with clinical outcome for patients treated with EGFR-TKIs has been reported [40]. Finally, no studies have evaluated polymorphisms in *CYP1A1* and their associations with toxicity.

9 CONCLUSIONS

The influence of gene polymorphisms in the EGFR pathway on clinical outcome and toxicity has been extensively investigated in advanced NSCLC patients treated with EGFR-TKIs. The *EGFR*-216 polymorphism have reported a strong association between response and/or survival to EGFR-TKIs in Caucasian population. Similarly, the positive effect of EGFR-CA repeats polymorphisms on survival of advanced NSCLC patients treated with EGFR-TKIs have been confirmed in both Caucasian and Asian population. The influence on toxicity of the -216, -191, CA repeats, Arg497Lys and Asp994Asp polymorphisms in *EGFR* have also been confirmed both in Caucasian and Asian population.

Polymorphisms in *AKT* (rs1130214 and rs1130233) and *SMAD3* (rs6494633, rs11071938 and rs11632964) have been associated with survival in advanced NSCLC patients treated with EGFR-TKIs. However, data come from a limited number of studies and need to be confirmed.

Finally, polymorphisms in genes coding proteins of the membrane transporters and cytochrome P450 enzymes have been less extensively investigated. There are few studies with small samples, which complicated the generalization of their role in EGFR-TKIs treatment.

In summary, polymorphisms in genes most extensively studied such as *EGFR* are promising biomarkers for the selection of treatment and follow-up of NSCLC patients. In clinical practice, *EGFR* polymorphisms may serve as a useful source of information to predict those patients with better response, higher survival and lower toxicity. Therefore, these biomarkers could be a valuable tool for patient stratification. However, polymorphisms in *AKT*, *SMAD3*, *ABCB1*, *CYP3A4*, *CYP3A5*, *CYP2D6*, *CYP1A1* genes need further examination in larger samples (stratified by gender, age and smoking status) and longer follow up.

10 REFERENCES

- 1 Siegel RL, Miller KD, Jemal A. Cancer statistics, 2017. CA: a cancer journal for clinicians 2017;67:7-30.
- Edge S, Byrd, D.R., Compton, C.C., Fritz, A.G., Greene, F.L., Trotti, A. Ajcc cancer staging manual, 7th ed., 2010.
- Herbst RS HJ, Lippman SM. Lung cancer. N Engl J Med 2008;359:1367-1380.
- 4 Molina JR, Yang P, Cassivi SD, Schild SE, Adjei AA. Non-small cell lung cancer: Epidemiology, risk factors, treatment, and survivorship. Mayo Clin Proc 2008;83:584-594.
- Ettinger DS, Wood DE, Akerley W, Bazhenova LA, Borghaei H, Camidge DR, Cheney RT, Chirieac LR, D'Amico TA, Dilling TJ, Dobelbower MC, Govindan R, Hennon M, Horn L, Jahan TM, Komaki R, Lackner RP, Lanuti M, Lilenbaum R, Lin J, Loo BW, Jr., Martins R, Otterson GA, Patel JD, Pisters KM, Reckamp K, Riely GJ, Schild SE, Shapiro TA, Sharma N, Stevenson J, Swanson SJ, Tauer K, Yang SC, Gregory K, Hughes M. Nccn guidelines insights: Non-small cell lung cancer, version 4.2016. Journal of the National Comprehensive Cancer Network: JNCCN 2016;14:255-264.
- Maemondo M, Inoue A, Kobayashi K, Sugawara S, Oizumi S, Isobe H, Gemma A, Harada M, Yoshizawa H, Kinoshita I, Fujita Y, Okinaga S, Hirano H, Yoshimori K, Harada T, Ogura T, Ando M, Miyazawa H, Tanaka T, Saijo Y, Hagiwara K, Morita S, Nukiwa T. Gefitinib or chemotherapy for non-small-cell lung cancer with mutated egfr. The New England journal of medicine 2010;362:2380-2388.
- Mitsudomi T, Morita S, Yatabe Y, Negoro S, Okamoto I, Tsurutani J, Seto T, Satouchi M, Tada H, Hirashima T, Asami K, Katakami N, Takada M, Yoshioka H, Shibata K, Kudoh S, Shimizu E, Saito H, Toyooka S, Nakagawa K, Fukuoka M. Gefitinib versus cisplatin plus docetaxel in patients with non-small-cell lung cancer harbouring mutations of the epidermal growth factor receptor (wjtog3405): An open label, randomised phase 3 trial. The lancet oncology 2010;11:121-128.
- Rosell R, Carcereny E, Gervais R, Vergnenegre A, Massuti B, Felip E, Palmero R, Garcia-Gomez R, Pallares C, Sanchez JM, Porta R, Cobo M, Garrido P, Longo F, Moran T, Insa A, De Marinis F, Corre R, Bover I, Illiano A, Dansin E, de Castro J, Milella M, Reguart N, Altavilla G, Jimenez U, Provencio M, Moreno MA, Terrasa J, Munoz-Langa J, Valdivia J, Isla D, Domine M, Molinier O, Mazieres J, Baize N, Garcia-Campelo R, Robinet G, Rodriguez-Abreu D, Lopez-Vivanco G, Gebbia V, Ferrera-Delgado L, Bombaron P, Bernabe R, Bearz A, Artal A, Cortesi E, Rolfo C, Sanchez-Ronco M, Drozdowskyj A, Queralt C, de Aguirre I, Ramirez JL, Sanchez JJ, Molina MA, Taron M, Paz-Ares L. Erlotinib versus standard chemotherapy as first-line treatment for european patients with advanced egfr mutation-positive non-small-cell lung cancer (eurtac): A multicentre, open-label, randomised phase 3 trial. Lancet Oncol 2012;13:239-246.
- 9 Sequist LV, Yang JC, Yamamoto N, O'Byrne K, Hirsh V, Mok T, Geater SL, Orlov S, Tsai CM, Boyer M, Su WC, Bennouna J, Kato T, Gorbunova V, Lee KH, Shah R, Massey D, Zazulina V, Shahidi M, Schuler M. Phase iii study of afatinib or cisplatin plus pemetrexed in patients with metastatic lung adenocarcinoma with egfr mutations. J Clin Oncol 2013;31:3327-3334.
- Bollinger MK, Agnew AS, Mascara GP. Osimertinib: A third-generation tyrosine kinase inhibitor for treatment of epidermal growth factor receptor-mutated non-small cell lung cancer with the acquired thr790met mutation. Journal of oncology pharmacy practice: official publication of the International Society of Oncology Pharmacy Practitioners 2017:1078155217712401.
- 11 Yarden Y, Schlessinger J. Self-phosphorylation of epidermal growth factor receptor: Evidence for a model of intermolecular allosteric activation. Biochemistry 1987;26:1434-1442.

- Lax I, Bellot F, Howk R, Ullrich A, Givol D, Schlessinger J. Functional analysis of the ligand binding site of egf-receptor utilizing chimeric chicken/human receptor molecules. The EMBO Journal 1989;8:421-427.
- 13 Katz M, Amit I, Yarden Y. Regulation of mapks by growth factors and receptor tyrosine kinases. Biochimica et biophysica acta 2007;1773:1161-1176.
- Pal SK, Figlin RA, Reckamp K. Targeted therapies for non-small cell lung cancer: An evolving landscape. Molecular cancer therapeutics 2010;9:1931-1944.
- Seshacharyulu P, Ponnusamy MP, Haridas D, Jain M, Ganti AK, Batra SK. Targeting the egfr signaling pathway in cancer therapy. Expert opinion on therapeutic targets 2012;16:15-31.
- Goss G, Tsai CM, Shepherd FA, Bazhenova L, Lee JS, Chang GC, Crino L, Satouchi M, Chu Q, Hida T, Han JY, Juan O, Dunphy F, Nishio M, Kang JH, Majem M, Mann H, Cantarini M, Ghiorghiu S, Mitsudomi T. Osimertinib for pretreated egfr thr790met-positive advanced non-small-cell lung cancer (aura2): A multicentre, open-label, single-arm, phase 2 study. The lancet oncology 2016;17:1643-1652.
- 17 Cheng L, Alexander RE, Maclennan GT, Cummings OW, Montironi R, Lopez-Beltran A, Cramer HM, Davidson DD, Zhang S. Molecular pathology of lung cancer: Key to personalized medicine. Modern pathology: an official journal of the United States and Canadian Academy of Pathology, Inc 2012;25:347-369.
- García-Foncillas J, Garrido P, Gómez J, Palacios J, Tarón M. Recomendaciones para la determinación de las mutaciones del gen egfr en el carcinoma de pulmón no microcítico. Revista Española de Patología 2011;44:17-31.
- 19 Reungwetwattana T, Weroha SJ, Molina JR. Oncogenic pathways, molecularly targeted therapies, and highlighted clinical trials in non-small-cell lung cancer (nsclc). Clinical lung cancer 2012;13:252-266.
- Rosell R, Moran T, Queralt C, Porta R, Cardenal F, Camps C, Majem M, Lopez-Vivanco G, Isla D, Provencio M, Insa A, Massuti B, Gonzalez-Larriba JL, Paz-Ares L, Bover I, Garcia-Campelo R, Moreno MA, Catot S, Rolfo C, Reguart N, Palmero R, Sanchez JM, Bastus R, Mayo C, Bertran-Alamillo J, Molina MA, Sanchez JJ, Taron M. Screening for epidermal growth factor receptor mutations in lung cancer. The New England journal of medicine 2009;361:958-967.
- Yasuda H, Kobayashi S, Costa DB. Egfr exon 20 insertion mutations in non-small-cell lung cancer: Preclinical data and clinical implications. The lancet oncology 2012;13:e23-e31.
- Thongprasert S, Duffield E, Saijo N, Wu YL, Yang JC, Chu DT, Liao M, Chen YM, Kuo HP, Negoro S, Lam KC, Armour A, Magill P, Fukuoka M. Health-related quality-of-life in a randomized phase iii first-line study of gefitinib versus carboplatin/paclitaxel in clinically selected patients from asia with advanced nsclc (ipass). J Thorac Oncol 2011;6:1872-1880.
- 23 Erdem L, Giovannetti E, Leon LG, Honeywell R, Peters GJ. Polymorphisms to predict outcome to the tyrosine kinase inhibitors gefitinib, erlotinib, sorafenib and sunitinib. Current topics in medicinal chemistry 2012;12:1649-1659.
- Zhang X, Fan J, Li Y, Lin S, Shu P, Ni J, Qin S, Zhang Z. Polymorphisms in epidermal growth factor receptor (egfr) and akt1 as possible predictors of clinical outcome in advanced non-small-cell lung cancer patients treated with egfr tyrosine kinase inhibitors. Tumour biology: the journal of the International Society for Oncodevelopmental Biology and Medicine 2016;37:1061-1069.
- Giovannetti E, Zucali PA, Peters GJ, Cortesi F, D'Incecco A, Smit EF, Falcone A, Burgers JA, Santoro A, Danesi R, Giaccone G, Tibaldi C. Association of polymorphisms in akt1 and egfr with clinical outcome and toxicity in non-small cell lung cancer patients treated with gefitinib. Molecular cancer therapeutics 2010;9:581-593.

- Massagué J. Tgfβ signalling in context. Nature reviews Molecular cell biology 2012;13:616-630.
- Hamada A SJ, Saeki S, Iwamoto N, Inaba M, Ushijima S, Urata M, Kishi H, Fujii S, Semba H, Kashiwabara K, Tsubata Y, Kai Y, Isobe T, Kohrogi H, Saito H. Association of abcb1 polymorphisms with erlotinib pharmacokinetics and toxicity in japanese patients with non-small-cell lung cancer. Pharmacogenomics 2012;13:615-624.
- Liu W, Innocenti F, Wu MH, Desai AA, Dolan ME, Cook EH, Jr., Ratain MJ. A functional common polymorphism in a sp1 recognition site of the epidermal growth factor receptor gene promoter. Cancer research 2005;65:46-53.
- 29 Gebhardt F, Zanker KS, Brandt B. Modulation of epidermal growth factor receptor gene transcription by a polymorphic dinucleotide repeat in intron 1. The Journal of biological chemistry 1999;274:13176-13180.
- Gregorc V, Hidalgo M, Spreafico A, Cusatis G, Ludovini V, Ingersoll RG, Marsh S, Steinberg SM, Vigano MG, Ghio D, Villa E, Sparreboom A, Baker SD. Germline polymorphisms in egfr and survival in patients with lung cancer receiving gefitinib. Clinical pharmacology and therapeutics 2008;83:477-484.
- Liu G, Gurubhagavatula S, Zhou W, Wang Z, Yeap BY, Asomaning K, Su L, Heist R, Lynch TJ, Christiani DC. Epidermal growth factor receptor polymorphisms and clinical outcomes in non-small-cell lung cancer patients treated with gefitinib. The pharmacogenomics journal 2008;8:129-138.
- Rudin CM, Liu W, Desai A, Karrison T, Jiang X, Janisch L, Das S, Ramirez J, Poonkuzhali B, Schuetz E, Fackenthal DL, Chen P, Armstrong DK, Brahmer JR, Fleming GF, Vokes EE, Carducci MA, Ratain MJ. Pharmacogenomic and pharmacokinetic determinants of erlotinib toxicity. J Clin Oncol 2008;26:1119-1127.
- Parmar S, Schumann C, Rudiger S, Boeck S, Heinemann V, Kachele V, Seeringer A, Paul T, Seufferlein T, Stingl JC. Pharmacogenetic predictors for egfr-inhibitor-associated skin toxicity. The pharmacogenomics journal 2013;13:181-188.
- Huang CL, Yang CH, Yeh KH, Hu FC, Chen KY, Shih JY, Lin ZZ, Yu CJ, Cheng AL, Yang PC. Egfr intron 1 dinucleotide repeat polymorphism is associated with the occurrence of skin rash with gefitinib treatment. Lung cancer 2009;64:346-351.
- Moriai T, Kobrin MS, Korc M. Cloning of a variant epidermal growth factor receptor. Biochemical and biophysical research communications 1993;191:1034-1039.
- Moriai T, Kobrin MS, Hope C, Speck L, Korc M. A variant epidermal growth factor receptor exhibits altered type alpha transforming growth factor binding and transmembrane signaling. Proceedings of the National Academy of Sciences of the United States of America 1994;91:10217-10221.
- Sasaki H, Okuda K, Takada M, Kawahara M, Kitahara N, Matsumura A, luchi K, Kawaguchi T, Kubo A, Endo K, Kawano O, Yukiue H, Yano M, Fujii Y. A novel egfr mutation d1012h and polymorphism at exon 25 in japanese lung cancer. Journal of cancer research and clinical oncology 2008;134:1371-1376.
- Sauna ZE, Kimchi-Sarfaty C, Ambudkar SV, Gottesman MM. Silent polymorphisms speak: How they affect pharmacogenomics and the treatment of cancer. Cancer research 2007;67:9609-9612.
- Sasaki H, Okuda K, Shimizu S, Takada M, Kawahara M, Kitahara N, Okumura M, Matsumura A, Iuchi K, Kawaguchi T, Kubo A, Kawano O, Yukiue H, Yano M, Fujii Y. Egfr r497k polymorphism is

- a favorable prognostic factor for advanced lung cancer. Journal of cancer research and clinical oncology 2009;135:313-318.
- Nie Q, Yang XN, An SJ, Zhang XC, Yang JJ, Zhong WZ, Liao RQ, Chen ZH, Su J, Xie Z, Wu YL. Cyp1a1*2a polymorphism as a predictor of clinical outcome in advanced lung cancer patients treated with egfr-tki and its combined effects with egfr intron 1 (ca)n polymorphism. European journal of cancer 2011;47:1962-1970.
- Winther-Larsen A, Nissen PH, Jakobsen KR, Demuth C, Sorensen BS, Meldgaard P. Genetic polymorphism in the epidermal growth factor receptor gene predicts outcome in advanced non-small cell lung cancer patients treated with erlotinib. Lung cancer 2015;90:314-320.
- Ma F, Sun T, Shi Y, Yu D, Tan W, Yang M, Wu C, Chu D, Sun Y, Xu B, Lin D. Polymorphisms of egfr predict clinical outcome in advanced non-small-cell lung cancer patients treated with gefitinib. Lung cancer 2009;66:114-119.
- Zhang L, Yuan X, Chen Y, Du XJ, Yu S, Yang M. Role of egfr snps in survival of advanced lung adenocarcinoma patients treated with gefitinib. Gene 2013;517:60-64.
- Masszi A, Kapus A. Smaddening complexity: The role of smad3 in epithelial-myofibroblast transition. Cells, tissues, organs 2011;193:41-52.
- Zhang L, Li QX, Wu HL, Lu X, Yang M, Yu SY, Yuan XL. Snps in the transforming growth factor-beta pathway as predictors of outcome in advanced lung adenocarcinoma with egfr mutations treated with gefitinib. Annals of oncology: official journal of the European Society for Medical Oncology / ESMO 2014;25:1584-1590.
- Agarwal S, Sane R, Gallardo JL, Ohlfest JR, Elmquist WF. Distribution of gefitinib to the brain is limited by p-glycoprotein (abcb1) and breast cancer resistance protein (abcg2)-mediated active efflux. The Journal of pharmacology and experimental therapeutics 2010;334:147-155.
- Kawamura K, Yamasaki T, Yui J, Hatori A, Konno F, Kumata K, Irie T, Fukumura T, Suzuki K, Kanno I, Zhang MR. In vivo evaluation of p-glycoprotein and breast cancer resistance protein modulation in the brain using [(11)c]gefitinib. Nuclear medicine and biology 2009;36:239-246.
- Imai Y, Nakane M, Kage K, Tsukahara S, Ishikawa E, Tsuruo T, Miki Y, Sugimoto Y. C421a polymorphism in the human breast cancer resistance protein gene is associated with low expression of q141k protein and low-level drug resistance. Molecular cancer therapeutics 2002;1:611-616.
- Kondo C, Suzuki H, Itoda M, Ozawa S, Sawada J, Kobayashi D, Ieiri I, Mine K, Ohtsubo K, Sugiyama Y. Functional analysis of snps variants of bcrp/abcg2. Pharmaceutical research 2004;21:1895-1903.
- Kobayashi D, leiri I, Hirota T, Takane H, Maegawa S, Kigawa J, Suzuki H, Nanba E, Oshimura M, Terakawa N, Otsubo K, Mine K, Sugiyama Y. Functional assessment of abcg2 (bcrp) gene polymorphisms to protein expression in human placenta. Drug metabolism and disposition: the biological fate of chemicals 2005;33:94-101.
- Fung KL, Gottesman MM. A synonymous polymorphism in a common mdr1 (abcb1) haplotype shapes protein function. Biochimica et biophysica acta 2009;1794:860-871.
- Marzolini C, Paus E, Buclin T, Kim RB. Polymorphisms in human mdr1 (p-glycoprotein): Recent advances and clinical relevance. Clinical pharmacology and therapeutics 2004;75:13-33.
- Sakaeda T, Nakamura T, Okumura K. Pharmacogenetics of mdr1 and its impact on the pharmacokinetics and pharmacodynamics of drugs. Pharmacogenomics 2003;4:397-410.
- Hoffmeyer S, Burk O, von Richter O, Arnold HP, Brockmoller J, Johne A, Cascorbi I, Gerloff T, Roots I, Eichelbaum M, Brinkmann U. Functional polymorphisms of the human multidrug-

resistance gene: Multiple sequence variations and correlation of one allele with p-glycoprotein expression and activity in vivo. Proceedings of the National Academy of Sciences of the United States of America 2000;97:3473-3478.

- Morisaki K, Robey RW, Ozvegy-Laczka C, Honjo Y, Polgar O, Steadman K, Sarkadi B, Bates SE. Single nucleotide polymorphisms modify the transporter activity of abcg2. Cancer chemotherapy and pharmacology 2005;56:161-172.
- Mizuarai S, Aozasa N, Kotani H. Single nucleotide polymorphisms result in impaired membrane localization and reduced atpase activity in multidrug transporter abcg2. International journal of cancer Journal international du cancer 2004;109:238-246.
- 57 Sakaeda T, Nakamura T, Okumura K. Mdr1 genotype-related pharmacokinetics and pharmacodynamics. Biol Pharm Bull 2002;25:1391-1400.
- Thiebaut F, Tsuruo T, Hamada H, Gottesman MM, Pastan I, Willingham MC. Cellular localization of the multidrug-resistance gene product p-glycoprotein in normal human tissues. Proc Natl Acad Sci U S A 1987;84:7735-7738.
- Tamura M KM, Horio M, Ando M, Saito H, Yamamoto M, Horio Y, Hasegawa Y. Genetic polymorphisms of the adenosine triphosphate-binding cassette transporters (abcg2, abcb1) and gefitinib toxicity. Nagoya J Med Sci 2012;74:133-140.
- Lemos C, Jansen G, Peters GJ. Drug transporters: Recent advances concerning bcrp and tyrosine kinase inhibitors. British journal of cancer 2008;98:857-862.
- 61 Li J, Cusatis G, Brahmer J, Sparreboom A, Robey RW, Bates SE, Hidalgo M, Baker SD. Association of variant abcg2 and the pharmacokinetics of epidermal growth factor receptor tyrosine kinase inhibitors in cancer patients. Cancer biology & therapy 2007;6:432-438.
- Akasaka K, Kaburagi T, Yasuda S, Ohmori K, Abe K, Sagara H, Ueda Y, Nagao K, Imura J, Imai Y. Impact of functional abcg2 polymorphisms on the adverse effects of gefitinib in japanese patients with non-small-cell lung cancer. Cancer chemotherapy and pharmacology 2010;66:691-698.
- Lemos C GE, Zucali PA, Assaraf YG, Scheffer GL, van der Straaten T, D'Incecco A, Falcone A, Guchelaar HJ, Danesi R, Santoro A, Giaccone G, Tibaldi C, Peters GJ. Impact of abcg2 polymorphisms on the clinical outcome and toxicity of gefitinib in non-small-cell lung cancer patients. Pharmacogenomics 2011;12:159-170.
- McKillop D, McCormick AD, Millar A, Miles GS, Phillips PJ, Hutchison M. Cytochrome p450-dependent metabolism of gefitinib. Xenobiotica; the fate of foreign compounds in biological systems 2005;35:39-50.
- Li J, Zhao M, He P, Hidalgo M, Baker SD. Differential metabolism of gefitinib and erlotinib by human cytochrome p450 enzymes. Clinical cancer research: an official journal of the American Association for Cancer Research 2007;13:3731-3737.
- Scheffler M, Di Gion P, Doroshyenko O, Wolf J, Fuhr U. Clinical pharmacokinetics of tyrosine kinase inhibitors: Focus on 4-anilinoquinazolines. Clinical pharmacokinetics 2011;50:371-403.
- Swaisland HC, Cantarini MV, Fuhr R, Holt A. Exploring the relationship between expression of cytochrome p450 enzymes and gefitinib pharmacokinetics. Clinical pharmacokinetics 2006;45:633-644.
- 68 Kobayashi H, Sato K, Niioka T, Miura H, Ito H, Miura M. Relationship among gefitinib exposure, polymorphisms of its metabolizing enzymes and transporters, and side effects in japanese patients with non-small-cell lung cancer. Clinical lung cancer 2015;16:274-281.

- Takimoto T, Kijima T, Otani Y, Nonen S, Namba Y, Mori M, Yokota S, Minami S, Komuta K, Uchida J, Imamura F, Furukawa M, Tsuruta N, Fujio Y, Azuma J, Tachibana I, Kumanogoh A. Polymorphisms of cyp2d6 gene and gefitinib-induced hepatotoxicity. Clinical lung cancer 2013;14:502-507.
- 70 Human cytochrome p450 (cyp) allele nomenclature committee. 2017.
- Kiyohara C, Hirohata T, Inutsuka S. The relationship between aryl hydrocarbon hydroxylase and polymorphisms of the cyp1a1 gene. Japanese journal of cancer research: Gann 1996;87:18-24.
- Kiyohara C, Nakanishi Y, Inutsuka S, Takayama K, Hara N, Motohiro A, Tanaka K, Kono S, Hirohata T. The relationship between cyp1a1 aryl hydrocarbon hydroxylase activity and lung cancer in a japanese population. Pharmacogenetics 1998;8:315-323.